

Author's response to reviews

Title: Improved Performance of Xpert® MTB/RIF Assay on Sputum Sediment Samples Obtained from Presumptive Pulmonary Tuberculosis Cases at Kibong'oto Infectious Diseases Hospital, Tanzania

Authors:

Peter Mbelele (mbelelepeter@yahoo.com)

Said Aboud (aboudsaid@yahoo.com)

Stellah Mpagama (sempagama@yahoo.com)

Mecky Matee (mateemecky@yahoo.com)

Version: 2 Date: 29 Nov 2017

Author's response to reviews:

The authors' response letter has been included as a supplementary file